Are equity market anomalies disappearing? Evidence from the U.K.

John Cotter and Niall McGeever University College Dublin

Abstract

We study the persistence over time of a set of well-known equity market anomalies in the cross-section of U.K. stocks. This market provides an excellent setting in which to study anomaly persistence as it is among the most developed and liquid in the world. We find strong evidence of diminished statistical significance for the return reversal effect, the momentum effect, and a number of other well-documented anomalies. These results hold for both portfolio sorting and Fama-MacBeth regression analyses and are robust to the use of alternative methods of risk adjustment and regression model specifications. Our findings are consistent with improvements in market efficiency over time with respect to well-known anomaly variables.

Keywords: Anomalies, Asset Pricing, Market Efficiency JEL Classification: G10, G12

Smurfit Graduate School of Business, University College Dublin, Blackrock, Co. Dublin, Ireland. We thank Michael Brennan, Gregory Connor, John J. McConnell, René Stulz, Avanidhar Subrahmanyam, and seminar participants at University College Dublin and Xiamen University for helpful comments and suggestions. We also thank Andrea Frazzini for making risk factor data available. We gratefully acknowledge the support of Science Foundation Ireland under grant number 08/SRC/FMC1389.

1 Introduction

Why do average returns vary across stocks? Standard rational-expectations theories of investor behaviour propose that a stocks risk determines its return. A number of celebrated models have sought to quantify the risk of a stock and relate it its expected return. The Capital Asset Pricing Model (CAPM) of Sharpe (1964), Lintner (1965), and Black (1972) is perhaps the most widely known of these models. Cochrane (2005) reviews a number of other seminal works including the Intertemporal CAPM of Merton (1973), the Arbitrage Pricing Theory of Ross (1976), and the Consumption CAPM of Lucas (1978) and Breeden (1979).

Anomalies are patterns in historical returns which are not anticipated by established risk-based models. A large body of evidence demonstrates that particular categories of stocks earn average returns which deviate significantly from the expectation of a specified risk-based model. Early contributions to this literature include the observation that firm size is negatively related to returns (Banz, 1981) and that value measures are positively related to returns (Basu, 1977). The documentation of robust anomalies indicates that expected returns vary according to non-risk firm characteristics or that the benchmark model is failing to capture some important aspect of risk.

Fama and French (1992) examine a large set of anomalies and summarise the main empirical failings of the Sharpe-Lintner-Black CAPM. They find that the size and value effects are the most difficult to reconcile with the model. In response, Fama and French (1993) develop size- and value-based risk factors and incorporate them into a model of expected returns. This model has supplanted the CAPM as the standard method for risk adjustment in the empirical asset pricing literature. Indeed, an additional fourth "momentum" factor - due to Carhart (1997) - is often added to this model.

The modern anomalies literature continues to grow. Subrahmanyam (2010) estimates that are at least 50 variables that have been related to the cross-section of expected returns. Establishing the economic significance of these empirical phenomena is an important goal for researchers in asset pricing.

A number of recent papers have made important contributions to the equity market anomalies literature. Harvey et al. (2015) question the statistical significance of many reported anomalies due to concern over data mining by researchers. They argue anomaly discovery rates rise spuriously as researchers investigate the same datasets repeatedly.

Other recent studies have considered the changing behaviour of more

robust anomalies over time. McLean and Pontiff (2015) present evidence suggesting that academic publications play a significant role in the disappearance of anomalies. They argue that empirical discoveries in the academic literature prompt investors to the exploit profitable anomalies.

Chordia et al. (2014), on the other hand, provide evidence of anomalies diminishing more gradually over time. They too interpret these results as improvement in the informational efficiency of U.S. stocks with respect to anomalous variables. They argue that this change is driven by improved liquidity and the rise in institutional trading and arbitrage activity. Similarly to McLean and Pontiff (2015), the authors argue that arbitrage activity such as hedge fund investing - has driven the decline in anomaly strength observed in the U.S. stock market.

Akbas et al. (2015) provide additional evidence on this front by directly linking changes in mutual fund flows ("dumb money") and hedge fund flows ("smart money") to variations in anomaly strength. The improvement of market liquidity and the increased level institutional trading has not been confined to the U.S. An interesting and unexplored question is whether these trends are evident in other developed markets.

In this paper, we examine the persistence over time of a set of well-known equity market anomalies in the cross-section of U.K. stocks. This market provides an excellent setting in which to study anomaly persistence. First, it is among the most developed and liquid stock markets in the world. Second, trading costs for U.K. stocks have fallen substantially over recent years (Brogaard et al., 2014). Third, the most robust anomalies in the literature are also evident in past U.K. returns.

We examine nine anomalies from the literature that have been robust across time and across international markets. The firm characteristic anomalies we consider are accruals (Sloan, 1996), asset growth Cooper et al. (2008), book-to-market ratio (Fama and French, 1992), new issuances of equity (Pontiff and Woodgate, 2008), profitability (Fama and French, 2006), return reversal (Jegadeesh, 1990), momentum (Jegadeesh and Titman, 1993), size (Banz, 1981), and stock turnover (Datar et al., 1998).

We find a general decline in the significance of anomalies in the U.K. market from 1990 to 2013. Many of the anomalies we examine are absent from the data in recent years. Perhaps most notably, the well-known momentum effect is not evident in the cross-section of U.K. stocks from 2002 to 2013. The profitability and turnover effects, however, remain rather robust throughout the sample. We find that the accruals anomaly is also not robust to the inclusion of additional firm characteristics in our Fama-MacBeth

regressions. We also find no evidence of a new equity issuance anomaly in our sample.

In addition, we examine the monthly time series of Fama-MacBeth coefficients for each anomaly variable. We find evidence of significant time trends in these series. In most cases, the coefficient series trend towards zero over time. These findings are consistent with declining hedge portfolio strategy returns and summarised in our main Fama-MacBeth regression results; the anomalies in our sample have generally diminished in strength or disappeared.

Our results are consistent with an improvement in market efficiency with respect to well-known anomaly variables. The significance of publicly available anomaly variables to explain differences in returns across stocks has diminished substantially in recent years. The returns on anomaly-based portfolios have in most cases fallen substantially. Our paper makes a clear contribution to the U.K. asset pricing literature, but also provides insight into the issue of anomaly persistence which is relevant to investors more generally.

The remainder of the paper is organised as follows. Section 2 provides a review of the literature on equity market anomalies. Section 3 provides a discussion of the anomalies we examine in this paper. Section 4 presents the details of our data sample. Section 5 reviews our research methodology. Section 6 presents the results of our analysis and provides a discussion of their significance. Section 7 concludes.

2 Interpreting Anomalies

In this section, we review the major contributions to the literature on equity market anomalies. We frame our discussion around the principal question in the literature - what is the economic significance of an observed anomaly? We begin by illustrating the process of interpreting an anomaly in Figure 1. This simple tree structure summarises the vast anomalies literature; all contributions to the literature fit in at one or more stages of the tree.

<Figure 1>

The first stage of the process is to determine the statistical significance of an anomaly. We must consider the possibility that a documented anomaly is a spurious result arrived at by chance or by extensive data mining by

researchers.

The second stage of the process is to consider whether an anomaly results from the use of an inappropriate model of expected returns. Put simply, it could be that we are ignoring some important aspect of a stock's risk. "Abnormal returns" may disappear when we use an alternative model of expected returns.

The third stage of the process is to determine the viability of exploiting anomalous returns with a trading strategy. It may not be possible to realize "paper profits" from an anomaly due to transaction costs, trading restrictions, or more complex limits to arbitrage (Gromb and Vayanos, 2010).

We review contributions to the literature for each of these three stages in turn:

2.1 Statistical significance

Are anomalies truly statistically significant? This is the first question we must ask in our analysis of these phenomena. We know that investors and researchers search historical returns for patterns at odds with conventional theories. Does this behaviour make the discovery of spurious anomalies more common?

Data mining is the practice of repeatedly using the same dataset to test many different hypotheses. The likelihood of identifying a spurious anomaly - one with no economic significance - increases with the number of new variables we investigate. This is a first-order concern in anomalies research as almost all empirical work in this literature is non-experimental. This issue should weigh on the minds of researchers visiting and revisiting historical databases.

The distortionary impact of data mining on the analysis of anomalies has been stressed repeatedly in the literature. Some researchers have tried to directly analyse the influence the practice has on reported results. Harvey et al. (2015) consider the impact data mining has on the rate of anomaly discovery in the cross-section of stock returns. They argue that many anomalies lose their significance when we account for the intensive search researchers have conducted. Similarly, Sullivan et al. (1999) find that data mining by researchers has generated many spuriously significant results in the calendar anomalies literature.

This data mining problem may be aggravated by the incentives aca-

demic researchers typically face. Novel results may lead to acceptance to conference programmes, journal publications, and promotion. This may bias the focus of academic discussion towards spurious results that challenge existing theory and away from unexciting results in line with expectations. This is sometimes termed the "file drawer" or "publication bias" problem in academia. Novel results are given prominence in discussion while uninteresting results are more likely to be consigned to an academic's "file drawer".

An important aspect of research on anomalies is the replication of results across different samples. Economically meaningful cross-sectional anomalies should not be confined to a particular sample period in a particular market. Fama (1991) makes this point clearly. He states that anomalies "warrant out-of-sample tests before being accepted as regularities that are likely to be present in future returns."

Out-of-sample replication is our main tool in establishing whether or not a result is truly statistically significant. This involves testing for the presence of an anomaly in other time periods and in alternative markets. Indeed, most robust anomalies in the literature - including those we consider in our analysis - were first documented in the U.S. equity market and then investigated further in international data.

2.2 Compensation for risk or inefficiency?

Anomalies are relative phenomena; to identify a deviation from the norm, we require a definition of the norm. Thus, robust anomalies may be evidence of abnormally high or low returns or simply evidence that we have not fully accounted for the given asset's risk. This "dual hypothesis" problem is one of the enduring messages from Fama (1970).

Still, the type of risk anomalies may reflect is generally not clear. Theorists have struggled to provide ex-post risk-based explanations for many wellknown anomalies. Fama and French (1993) famously incorporate size- and value-based factors into a risk-based model of expected returns, but concede that these two variables have "no special standing in asset-pricing theory." This approach has drawn some sharp criticism. For instance, Shleifer (2000) cites the loose theoretical foundation for these factors as a major problem for the model. Researchers have faced the same difficulty with other wellknown anomalies such as momentum (Jegadeesh and Titman, 1993) and the short-run return reversal effect (Jegadeesh, 1990).

The alternative explanation is that robust anomalies are evidence of market inefficiency. A market price is said to be "efficient" in the most general sense if, on average, no information can help us improve upon its estimate of an asset's value. Many behavioural financial economists argue that anomalies arise due to irrational trading on the part of investors. Traders may neglect relevant information and place too much emphasis on other signals. Importantly, anomaly variables can typically be observed easily by the public. Behavioural researchers generally argue that investor sentiment pushes prices (and expected returns) away the levels justified by firm fundamentals. Shiller (2000) provides a famous account of this position.

The existence and persistence of market inefficiencies depend on the two influences - investor sentiment and rational arbitrage (smart money"). We discuss these two issues below.

2.2.1 Investor Sentiment

The traditional approach of microeconomics and its application in finance has been to develop models of rational decision-making under resource constraints. Homo economicus is given a utility function to maximise and a set of resources with which to do so. An alternative literature has developed over the last few decades. Researchers in behavioural economics have drawn inspiration from the psychology literature and analysed real-world economic decision-making of individuals.

Behavioural economists have documented a large number of decisionmaking biases that are difficult to reconcile with traditional models of individual utility. Examples include loss aversion (Kahneman and Tversky, 1979) and money illusion (Shafir et al., 1997). The message from this literature is clear; people are not always dispassionate utility maximisers. The implications of this for individual outcomes and market dynamics is now central to the debate in most fields of economics including finance.

Importantly, broad studies of economic behaviour may not lead to appropriate conclusions about the behaviour of investors. Direct participants in the stock market make up a relatively small subset of society. Wealth is a major factor in determining participation rates (Mankiw and Zeldes, 1991). Other important factors that researchers have highlighted include financial literacy (van Rooij et al., 2011), IQ (Grinblatt et al., 2011), and sociability (Hong et al., 2004). These findings highlight the importance of developing an understanding of the economic behaviour of investors as distinct from individuals in general.

Researchers in behavioural finance have found that investors suffer from many behavioural biases. Barberis and Thaler (2003) provide an excellent survey of these findings. For instance, investors are often over-confident about their ability to predict future market trends. Investors also attribute successful investments to their skill and attribute failures to bad luck.

Importantly, Kumar and Lee (2006) demonstrate that these biases have the potential to influence market prices. They show that the trading decisions of individual investors are systematically correlated. That is, individual investors consistently behave like other individual investors. The implication of this result is that investor sentiment must be counterbalanced by the actions of rational arbitrageurs. In the absence of such action, investor sentiment will support the persistence of market anomalies.

2.2.2 Smart money

The claim that anomalies are the result of psychological biases among investors is a contentious issue in finance. Proponents of market efficiency typically argue that competition for information in capital markets will result in irrational traders losing money. They argue that sophisticated investors -"smart money"- will eliminate market anomalies generated by investor sentiment among irrational traders. Friedman (1953) was perhaps the first to lay out this argument. Shleifer (2000) states that "the theoretical case for efficient markets rests on the effectiveness of such arbitrage."

A common argument in the literature is that the discovery of a profitable anomaly - an inefficiency - will lead to its destruction. Smart investors should, it is argued, exploit the abnormal returns on offer by constructing portfolios which capture these returns. Anecdotal evidence abounds of academic discoveries informing investor trading. More formally, McLean and Pontiff (2015) argue that the publication of academic papers directly leads to the elimination of anomalies.

Chordia et al. (2014), on the other hand, provide evidence of anomalies gradually diminishing over time. They link this to dramatic improvements in market liquidity and the growth in arbitrage activity in recent years. Akbas et al. (2015) go further and present evidence suggesting that aggregate fund flows are related to variations in the magnitude of anomalies. Mutual fund flows (a proxy for "dumb money") often result in anomalies becoming more pronounced and hedge fund flows (a proxy for "smart money") lead to anomaly attenuation.

The Adaptive Markets Hypothesis (Lo, 2004) presents an alternative perspective on these results which is rooted in evolutionary biology. The efficiency of a market in this context will depend on the intensity of competition among investors for exploitable anomalies. Intense competition will lead to the elimination of profitable opportunities and a decline in the population of traders. The theory anticipates cycles in the level of profitability anomalies offer in response to changing market conditions and the number of competing investors. Roll (1994) and Stambaugh (2014) explore similar ideas.

2.3 Exploitability

Can savvy investors profit from market anomalies? This is the subject of a growing literature on the limits to arbitrage. Theoretical contributions to this literature have demonstrated that arbitrageurs may be unable or unwilling to exploit anomalies for a variety of reasons.

Gromb and Vayanos (2010) describe this literature as having two branches. The first branch deals with fundamental limits to arbitrage. Trading costs are perhaps the most basic difficulty that investors face when trading on an anomaly. Though an anomaly may be persistent and may not be explained by variation in risk across stocks, investors may choose not to trade on it if it is overly costly to do so.

Another issue for arbitrageurs is that short selling may be restricted by government or stock exchange policy. In addition, position limits may prevent fund managers from constructing portfolios they see as optimal for capturing anomaly profits.

The second branch of this literature deals with non-fundamental limits to arbitrage. De Long et al. (1990) develop a model in which arbitrageurs face "noise trader risk". That is, the risk arbitrageurs face of irrational traders sustaining or intensifying anomalies after the arbitrageur has taken a position. Alternatively, Shleifer and Vishny (1997) develop a model in which an arbitrageur is concerned about their clients withdrawing funds in the event of temporarily negative performance. This incentivises the arbitrageur not to trade on anomalies.

Coordination failure among arbitrageurs is another potential problem for investors seeking to exploit anomalies. Abreu and Brunnermeier (2002) and Stein (2009) develop models in which arbitrageurs are unaware of the behaviour of other arbitrageurs. Trades can become "crowded" as many institutional traders try to exploit an anomaly. Rather than eliminating anomalies and driving prices closer to their fundamental value, arbitrageurs in this setting may generate substantial instability (at least in the short run). The main implication of this literature is that well-known market anomalies may persist over time. These models suggests that cross-sectional equity market anomalies - the subject of our study - may persist even in the presence of knowledgeable arbitrageurs.

3 Anomalies

In our study, we examine a set of well-known cross-sectional anomalies which have generated much discussion in the literature. These anomalies have shown to be robust across different samples and have proven difficult for theorists to incorporate into risk-based models of expected return. Table 1 lists the variables.

<Table 1>

Sloan (1996) shows that high levels of accounting accruals are associated with low average returns. He posits that investors do not fully appreciate the information that is contained in the components of firms' earnings. Cooper et al. (2008) find a negative relationship between the rate of firm asset growth and average stock returns. These authors argue that biased decision-making among investors is the best explanation for this anomaly. Specifically, investors may be naively extrapolating asset growth rates of firms.

Pontiff and Woodgate (2008) report a negative relationship between firm equity issuance and average returns. This result suggests that firm managers conduct equity issuances and share repurchases opportunistically. Again, these authors argue that their results are difficult to reconcile with riskbased models.

Fama and French (1992) present evidence that firm book equity to market equity ratios are positively related to average returns. The authors hypothesise that distress risk for firms might drive this relationship. Cochrane (2005) notes that this interpretation is not fully consistent with the current evidence.

Jegadeesh (1990) documents a short-run reversal effect in stock returns. That is, stock returns in a given month are negatively related to returns in the previous month. None of the risk-based models he considers can account for this predictability. Fama and French (2006) report a positive relationship between firm profitability and expected returns. They link this phenomenon to valuation theories of the firm. Jegadeesh and Titman (1993) report a "momentum" effect in stock returns whereby returns in a given month are positively related to performance in the last 3 to 12 months. They argue that over-reaction of investors to past performance may explain the pattern. Banz (1981) identifies a negative relationship between firm market capitalisation and average returns. He is unable to attribute this effect to a risk-based model of expected returns. Lastly, Datar et al. (1998) find a negative relationship between stock turnover and average returns. They interpret this result as evidence of low liquidity stocks earning higher average returns.

4 Data

In this section, we discuss the data we use in our analysis. We place particular emphasis on the process of enhancing the quality of the data through a careful screening process.

Our data sample consists of individual equity returns and firm characteristics for the cross-section of U.K. stocks from January 1990 to December 2013. We source these data from Thomson Reuters Datastream. We ensure that dead stocks are included to avoid a survivorship bias in our sample. In addition, we obtain market, size, value, and momentum risk factors published by AQR Capital Management.

Data quality is an important consideration for researchers studying asset pricing in international markets. Ince and Porter (2006) identify a number of systematic errors in individual equity databases. Thus, we proceed with caution. To counter these issues, Ince and Porter (2006) devise a set of screens which improve equity data reliability. Schmidt et al. (2014) build on this contribution and further improve these screening procedures.

We carefully clean our data using the static and dynamic screens outlined by Schmidt et al. (2014). This is the first paper to use these more refined procedures with U.K. data. We retain stocks that have a price history and a return history of at least 24 months, are identified by Datastream as "equity" securities, are the major security of the given firm, are listed in London, and are identified as U.K. companies.

We also remove securities - such as preference shares - which may have escaped the above filters by searching security extended names for the following terms: "pref", "prf", "%", "dupl", and "duplicate".

We next conduct an exploratory analysis of the data to identify suspect

outliers. More formally, we set a firm's market value equal to missing if the difference between it and unadjusted price times number of shares outstanding is larger than 0.5 in absolute value. We also set monthly returns equal to missing if they are greater than 990% or if either price term is greater than $\pounds1,000,000$.

The London Stock Exchange has historically contained many small firms. Researchers analysing this market have typically filtered out firms with very low stock prices. We follow Nagel (2002) and filter out stocks with nominal prices below £0.30. In addition, we remove securities which have unadjusted prices below the 5th percentile of the price distribution for the full sample as a "penny stock"-style filter.

Our final sample is comprised of 1,229 stocks. We believe our battery of static and dynamic screens improved the quality of our dataset.

5 Methodology

In this section, we discuss our methodological approach. We conduct two distinct analyses. First, we consider the returns generated by trading strategies designed around anomaly variables. In each month, we sort stocks into quintile portfolios based on the value of a given anomaly variable. We then calculate returns for a strategy which is long the portfolio with the highest values of the anomaly variable and short the portfolio with the lowest values of the anomaly variable. This provides a tangible measure of the economic magnitude of each anomaly in the form of monthly percentage trading returns.

We also consider the robustness of our hedge portfolio results to methodological changes. In particular, we vary the number of portfolios we sort stocks into each month and report results generated from monthly decile sorts.

Second, we use regression analysis to determine the statistical significance of our set of anomalies. We apply the modified version of the Fama and MacBeth (1973) two-step cross-sectional regression methodology proposed by Brennan et al. (1998).

We begin with a first-pass time series regression described by equation (1) to estimate sensitivies of individual firm returns to a set of L specified factors. We choose these factors from established models of expected return. This first pass regression gives us estimated betas with respect to L factors

for each firm j. We generate the results in the main body of the paper using the four-factor Fama and French (1993) and Carhart (1997) model.

$$\tilde{R}_{jt} = R_{Ft} + \sum_{k=1}^{L} \hat{\beta}_{jk} \tilde{f}_{kt}$$
(1)

The second-pass of the Fama-MacBeth procedure is to consider the significance of anomaly variables in explaining returns given what we know about the assets' betas. An important point to note is that measurement error is a well-known problem in firm betas (Miller and Scholes, 1972). Inclusion of mismeasured betas as independent variables in second-pass crosssectional regressions will result in biased and inconsistent Fama-MacBeth coefficient estimates.

The defining feature of our regression approach is the incorporation of the mismeasured first-pass betas into the dependent variable of the second-pass cross-sectional regression. We construct risk-adjusted firm returns \tilde{R}_{jt}^* for each firm j in equation (2).

$$\tilde{R}_{jt}^* = \tilde{R}_{jt} - R_{Ft} - \sum_{k=1}^L \hat{\beta}_{jk} \tilde{F}_{kt}$$
(2)

We next regress these returns on a set of specified anomaly variables in month-by-month cross-sectional regressions. We vary the set of anomalies to produce univariate and multivariate estimates c_m in equation (3).

$$\tilde{R}_{jt}^{*} = c_0 + \sum_{m=1}^{M} c_m Z_{mjt} + \tilde{e}'_{jt}$$
(3)

The use of risk-adjusted returns returns as the dependent variable in equation (3) allows for consistency in the anomaly variable coefficient estimates, but comes at the cost of greater inefficiency of these coefficient estimates (Hausman, 2001). Brennan et al. (1998) argue that this is, on balance, a good tradeoff and is an appropriate method for addressing this well-known errors-in-variables problem.

We then calculate Fama and MacBeth (1973) coefficients for each anomaly variable. These are the time series means of each variable's estimated premium. These coefficients equal zero under the null hypothesis that firm returns are determined by the set of L specified factors.

Finally, we look closer at the month-by-month Fama-MacBeth coefficient estimates for each anomaly variable. We examine the dynamics of these series over our sample period and conduct a trend analysis on them using simple t-tests.

6 Results

6.1 Hedge portfolio returns

We begin our analysis by estimating the returns of portfolios constructed with stocks sorted by anomaly variables. This analysis allows us to determine the returns investors could have enjoyed had they followed trading strategies based on these anomaly variables. As such, this approach provides a tangible measure of each anomaly's economic significance.

Each month, we sort our sample of stocks from highest to lowest value of a particular anomaly variable. We calculate the return on an equallyweighted portfolio which is long the quintile with the highest values of the variable and short the quintile with the lowest values of the variable. We then test whether the average monthly return of this portfolio is statistically different from 0%.

Table 2 presents the results of our quintile portfolio analysis. The first column of results shows average monthly portfolio returns over the full January 1990 to December 2013 sample period. The second and third columns show average portfolio returns for the first and second halves of the sample period.

<Table 2>

In the full sample, we find that five of the average portfolio returns are significantly different from zero. These are the asset growth, book-to-market, profitability, momentum, and turnover portfolios. These significant returns range from 0.836% per month for the book-to-market portfolio to 1.907% per month for the profitability portfolio.

The sub-sample results show that returns to these strategies have fallen for seven of the nine portfolios. The strong statistical significance of the asset growth, book-to-market, and momentum portfolios is sharply diminished in the second sub-sample. In the two cases where portfolio returns rose in absolute terms (the new issuance and size portfolios), neither return is statistically significant in the second sub-sample.

The returns of the profitability and turnover portfolios remain strongly significant in the second sub-sample. Nonetheless, the magnitude of their returns fell in both cases. The return on the profitability portfolio fell from 2.11% to 1.7%. The return on the turnover portfolio sees a more sizeable decline from 2.24% to 1.48%.

A valid concern with this analysis is that the number of portfolios we choose to sort stocks into is somewhat arbitrary. Why analyse quintiles and not sextiles or deciles? We seek to address this issue by varying the number of portfolios we sort stocks into as a robustness check. We find that our results are broadly insensitive to these changes and we reach the same qualitative conclusions regarding anomaly attenuation.

Table 3 presents the results of our hedge portfolio analysis repeated using decile portfolios. The first column of results again shows average monthly portfolio returns over the full January 1990 to December 2013 sample period. The second and third columns show average portfolio returns for the first and second halves of the sample period.

<Table 3>

In the full sample, we find that five of the average portfolio returns are significantly different from zero. These are the asset growth, book-to-market, profitability, momentum, and turnover portfolios. These significant returns range from 0.629% per month for the book-to-market portfolio to 2.260% per month for the turnover portfolio.

The sub-sample results show that returns to these strategies have fallen for six of the nine portfolios. The strong statistical significance of the asset growth, book-to-market, and momentum portfolios is sharply diminished in the second sub-sample.

Again, the profitability and turnover portfolios show strongly significant returns in the second sub-sample. The magnitude of these returns fell in both cases. The return on the profitability portfolio fell from 2.052% to 1.585%. The return on the turnover portfolio fell from 2.589% to 1.931%.

The returns for the accruals, new issuance, and size portfolios rose in absolute terms. However, none of these returns are adjudged to be statistically significant.

In summary, our hedge portfolio returns show a clear decline in the statistical significance of anomaly-based trading strategies. This holds even for the portfolios for which returns remain statistically significant in the second sub-sample.

6.2 Univariate Fama-MacBeth regressions

In this section, we examine the significance each of our nine anomaly variables using univariate Fama and MacBeth (1973)-style regressions. We regress risk-adjusted individual firm returns on each anomaly variable in turn following Brennan et al. (1998). This provides us with univariate Fama-MacBeth coefficients for each anomaly in each month of our sample. We adjust firm returns for risk using the four-factor Fama and French (1993) and Carhart (1997) model in all cases.

Table 4 presents the results of our univariate regression analysis. Each regression produces a one constant coefficient and one anomaly variable coefficient. We include results for the full January 1990 to December 2013 sample period and for the first and second halves period. The first sub-sample runs from January 1990 to December 2013. This is a common method in the literature to examine the persistence over time of anomalies. See, for example, Schwert (2003).

<Table 4>

The first column of Table 4 refers to the results from the full sample period. All but one of the nine anomaly variables are statistically significant at a 5% level for this period. Moreover, seven of these variables are significant at a 1% level. The accruals, book-to-market, profitability, return reversal, momentum, size, and turnover effects are all strongly significant.

The signs of the Fama-MacBeth coefficients generally correspond closely with the sorted long-short portfolio returns in Tables 2 and 3, though we find stronger statistical significance for accruals, return reversal, and size in our regression analysis.

The second and third columns of Table 4 the results of this univariate Fama-MacBeth regression analysis for the first and second halves of our sample. In the first sub-sample, seven anomaly variables are significant at a 5% level in the first sub-sample and five were significant at a 1% level. The coefficients for accruals, profitability, return reversal, momentum, and turnover are all positive and strongly significant.

The results for the second sub-sample shows a marked decline in significance for a number of variables. The Fama-MacBeth coefficients for accruals, book-to-market, return reversal, and momentum are all insignificant at a 5% level. The new issuance anomaly is again insignificant. The asset growth, profitability, and turnover coefficients remain significant, but have all fallen in magnitude. For example, the asset growth coefficient fell from 0.009 to 0.002.

The size anomaly is an exception to the observed trend of anomaly attenuation. It is the only variable we analyse that becomes significant in our second sub-sample. The coefficient for this variable is also positive. This is consistent with the positive (and insignificant) size portfolio return results reported in Tables 2 and 3. These results are surprising given that most estimates of the size effect in the literature are negative. We discuss this disparity further in Section 6.5.

In summary, we find that evidence of diminished statistical significance in the second sub-sample using univariate Fama-MacBeth regressions. Two exceptions to this trend are the profitability and turnover anomaly coefficients.

6.3 Multivariate Fama-MacBeth regressions

We next examine the significance of our anomaly variables using multivariate Fama and MacBeth (1973)-style regressions. We again follow Brennan et al. (1998), but now regress risk-adjusted individual firm returns on several anomaly variables. We vary the number of anomaly variables from two to the full specification of nine.

Table 5 presents the correlation matrix of our set of anomaly variables. We find that the correlations among the variables are quite low. Most are below 0.1 in absolute value. These are similar to the findings of Brennan et al. (1998). The highest correlation in our sample is 0.408 between size and turnover.

<Table 5>

Table 6 shows the results of our full-sample multivariate Fama-MacBeth regressions of firm returns on anomaly variables. We report a number of different specifications. With the correlation results of Table 5 in mind, we explore the impact of including one or both of the size and turnover variables in our final three specifications.

<Table 6>

The book-to-market, profitability, size, and turnover characteristics are all rather robust to alternative specifications. For instance, the largest of the profitability coefficients p-values is 0.003 and the coefficients rise as we add additional characteristics. The sign of all but sizes coefficients match those in the univariate regression results in Table 4.

The size coefficient is negative when we control for many other firm characteristics. This is in keeping with the original findings of Banz (1981), but conflicts with our univariate results in Table 4. The return-reversal coefficient is marginally significant for some specifications, but strongly significant in the final two specifications. The sign on the coefficient is also negative. This is what has been documented in previous studies (Jegadeesh, 1990) and is in contrast with the positive full sample univariate coefficient we report in Table 4.

We see that the coefficients of the other anomaly variables are quite sensitive to specification changes. The new issuances coefficient is significant for two specifications, but insignificant in all others. This is consistent with weak evidence for this effect in our univariate analysis in Table 4. The strong significance of the accruals coefficient disappears with the inclusion of additional anomaly variables. The asset growth coefficient is significant for most specifications, but insignificant at a 5% level in the final specification. The momentum coefficient also disappears with the inclusion of size and turnover.

Tables 7 and 8 present the sub-sample results for our multivariate Fama-MacBeth regressions. We use the same non-overlapping sub-samples as in the univariate analysis of Table 4. We again find that the significance of a number of anomaly variables has diminished in more recent years. For instance, six variables are significant at a 1% significance level in the final specification of Table 7 compared to three in Table 8.

<Table 7>

<Table 8>

The disappearance of the momentum anomaly is striking. The coefficient is highly significant in the first sub-sample and insignificant in all second sub-sample specifications it appears in. The inclusion of turnover in the regression makes this attenuation even clearer. The coefficient becomes negative in the final specification, but remains insignificant.

The diminishment of the momentum effect in our multivariate analysis is consistent with our univariate results in Table 4. Interestingly, the momentum effect also disappears in the second sub-sample when we adjust for risk without the use of a momentum factor (see Tables 14 and 17 in the Appendix). It is not the case that our momentum factor is sapping the statistical significance of the characteristic coefficient.

In summary, we again find evidence of weakening in the significance of anomaly coefficients using multivariate Fama-MacBeth regressions. Similarly to our univariate results, the profitability and turnover coefficients are rather robust over time and in different regression specifications.

6.4 Fama-MacBeth coefficient time trends

In this section, we examine the time series of multivariate Fama-MacBeth regression coefficients for each anomaly. The series correspond to a regression specification where we include all nine anomalies as independent variables. We use the four-factor Fama and French (1993) and Carhart (1997) model to risk-adjust individual firm returns.

We construct 60-month moving averages for the regression coefficients. The moving average calculation shortens the sample period by five years. This truncated sample runs from January 1995 to December 2013.

Figure 2 presents the time of Fama-MacBeth coefficient moving averages for each of the nine anomaly variables. Casual inspection shows that most series trend towards zero. There are two main exceptions. The series corresponding to the size and turnover variables appear to trend away from zero.

<Figure 2>

We conduct time-trend t-tests to produce more formal results. We present these results in Table 9. We find that most of the time trend coefficient have statistically significant time trends towards zero. Size and turnover are, as expected, the two exceptions. Both of these series have significant time trends away from zero. The Fama-MacBeth coefficients for size become more negative over time and the coefficients for turnover become more positive.

<Table 9>

Our results in Table 9 match closely the trends we see between the subsamples shown in Tables 7 and 8. The correspondence is unsurprising given that the statistics in these tables are drawn from the raw Fama-MacBeth coefficient time series.

6.5 Discussion

In this section, we provide a discussion of our results and place them in the context of the literature.

Our results show strong evidence of declining strength for most of the anomalies we examine. We find that prominent anomalies such as the momentum (Jegadeesh and Titman, 1993) and short-run return reversal (Jegadeesh, 1990) effects have diminished significantly in recent years.

Our results provide other important insights. For instance, the economic magnitudes of the anomalies we investigate can be substantial. For instance, the quintile portfolio profitability strategy in Table 2 has a full-sample return of 1.819% per month or approximately 22.8% per annum. This figure is even higher - 25.32% per annum - in the first sub-sample. Similarly, the momentum and turnover quintile portfolio strategy returns in the first sub-sample of Table 2 are of a similar magnitude. Chordia et al. (2014) find similar results using U.S. stocks. For example, they find that momentum portfolios earn between 17.27% and 22.58% per annum.

Our findings are robust to the use of alternative portfolio construction methods. The decile portfolio strategy returns in Table 3 are broadly similar to the quintile strategy returns in Table 2. The largest difference is for the turnover portfolio returns. This portfolio performs even better - 27.12% per annum - when constructed using decile portfolios. This improved performance is driven by the use stocks with more extreme characteristics.

The positive relationship we find between size and returns in Table 4 is perhaps puzzling in light of the historical evidence of a negative relationship between these variables. van Dijk (2011) shows that though the negative relationship between size and average returns has persisted in U.S. data, it has reversed several times. Indeed, many of these reversals have lasted a number of years. A potential explanation our result is a deviation of realised returns from expected returns (Elton, 1999). That is, the ex ante expectation of a negative relationship was not observed in our sample period by chance. It may be this ex ante expectation is still appropriate in future. Hou and van Dijk (2012) present evidence supporting this explanation. In any case, our analysis suggests that size continues to be related to average risk-adjusted stock returns.

We find strong evidence of anomaly significance in our univariate Fama-MacBeth regressions. These effects are again primarily evident in the first sub-sample. The most robust anomalies here are the profitability, size, and turnover effects. Another result of note is the significance of the asset growth anomaly in the second sub-sample.

We find that not all anomalies are robust to the inclusion of other variables in multivariate Fama-MacBeth specifications. For instance, the accruals coefficient becomes insignificant when we account for a wider set of variables. The asset growth anomaly coefficient is also sensitive to the addition of other anomaly variables. In Table 6, it becomes marginally insignificant when we account for firm size. This coefficient is also insignificant for many of the specifications in the second sub-sample in Table 8. The most robust anomalies are again the profitability, size, and turnover effects.

The results of our time trend analysis in Table 9 provide another perspective on the data. We find significant time trends toward zero for most of the anomaly variable coefficient series we examine. These findings are consistent with the conclusions we draw from the sub-sample Fama-MacBeth regression results in Tables 7 and 8.

What economic interpretation can we draw from our results? Firstly, we document out-of-sample persistence for many of the anomalies we examine. This result is inconsistent with the anomalies simply being the product of data mining by researchers. They appear to be robust in the sense of statistical significance.

Secondly, we also find that many of our anomalies are robust to the use of alternative models of risk-adjustment. It is possible that a richer model of expected returns could explain the anomalous variation in returns we report. However, these anomalies have proven difficult for theorists to reconcile with risk-based theoretical models.

Thirdly, our findings are consistent with improvements in market efficiency. The growth of arbitrage activity in recent years could explain the trends we report. In particular, hedge fund activity has grown substantially (Hanson and Sunderam, 2014; Stein, 2009). We have much anecdotal evidence that these funds target anomalies. Akbas et al. (2015) and McLean and Pontiff (2015) present more formal evidence showing that hedge funds do exploit anomalies. Boehmer and Kelley (2009) also report lower predictability in the returns of stocks with a higher proportion of institutional investors.

Our evidence is also consistent with the findings of recent studies that have looked at other aspects of pricing efficiency in financial markets. For example, Chaboud et al. (2014) find report a decrease in the number of triangular arbitrage opportunities in foreign exchange markets. Also, Jones and Pomorski (2013) find that short-run autocorrelations in stock returns have diminished in recent years.

An important point to note is that the decline in anomaly significance we observe may not continue indefinitely. Models in the limits to arbitrage literature predict that anomalies will emerge and persist if investor sentiment influences expected returns and a sufficiently large supply of arbitrage capital does not seek to correct this influence.

Long-Term Capital Management (LTCM) is an excellent example of the dangers facing hedge funds targeting market anomalies. Though the fund was initially very successful, Lowenstein (2000) documents how deteriorating market conditions led to the collapse of the fund. The experience of LTCM highlights the potential sensitivity of anomaly-based investment strategies to swings in the investor sentiment.

In summary, our results show a clear trend of diminishing significance for most of the anomalies we examine.

7 Conclusion

We find strong evidence of anomaly attenuation in the cross-section of U.K. stocks from 1990 to 2013. We confirm the existence of several well-known anomalies in this market, but show that the statistical significance of these anomalies has diminished markedly in recent years. This pattern holds for the momentum (Jegadeesh and Titman (1993)), book-to-market ratio (Fama and French (1992)), accruals (Sloan (1996)), and asset growth (Cooper et al. (2008)) effects. Our findings are robust to the use of the use of hedge portfolio strategies, Fama-MacBeth regressions, and various model specifications.

The profitability (Fama and French, 2006) and turnover (Datar et al., 1998) anomalies remain rather robust throughout our analysis. We do find that portfolio strategy returns based on these anomalies fell over our sample, but these returns remain statistically significant. This also holds true for Fama-MacBeth regression coefficients for these two anomalies. Furthermore, we document statistically significant time trends in Fama-MacBeth coefficient series for many of our anomaly variables. We find that most coefficient series trend towards zero over the sample period.

This paper provides insight into how stocks are priced in the U.K. market. It also adds to the body of evidence on the persistence of equity market anomalies over time. Our findings are consistent with improvements in market efficiency over time with respect to well-known anomaly variables. Further work on this topic might consider a direct link between institutional investment behaviour, measures of investor sentiment, and the strength of anomalies in the U.K. equity market.

References

- Abreu, Dilip, and Markus K. Brunnermeier, 2002, Synchronization risk and delayed arbitrage, *Journal of Financial Economics* 66, 341–360.
- Akbas, Ferhat, Will J. Armstrong, Sorin Sorescu, and Avanidhar Subrahmanyam, 2015, Smart money, dumb money, and equity return anomalies, *Journal of Financial Economics* 118, 355–382.
- Banz, Rolf W., 1981, The relationship between return and market value of common stocks, *Journal of Financial Economics* 9, 3–18.
- Barberis, Nicholas, and Richard Thaler, 2003, A survey of behavioral finance, in George M. Constantinides, Milton Harris, and René M. Stulz, eds., *Handbook of the Economics of Finance* (Elsevier, Amsterdam).
- Basu, Sanjoy, 1977, Investment performance of common stocks in relation to their price-earnings ratios: A test of the efficient market hypothesis, *Journal of Finance* 32, 663–682.
- Black, Fischer, 1972, Capital market equilibrium with restricted borrowing, Journal of Business 45, 444–455.
- Boehmer, Ekkehart, and Eric K. Kelley, 2009, Institutional investors and the informational efficiency of prices, *Review of Financial Studies* 22, 3563–3594.
- Breeden, Douglas T., 1979, An intertemporal asset pricing model with stochastic consumption and investment opportunities, *Journal of Finan*cial Economics 7, 265–296.
- Brennan, Michael J., Tarun Chordia, and Avanidhar Subrahmanyam, 1998, Alternative factor specifications, security characteristics, and the crosssection of expected stock returns, *Journal of Financial Economics* 49, 345–373.
- Brogaard, Jonathan, Terrence Hendershott, Stefan Hunt, and Carla Ysusi, 2014, High-frequency trading and the execution costs of institutional investors, *Financial Review* 49, 345–369.

- Carhart, Mark M., 1997, On persistence in mutual fund performance, Journal of Finance 52, 57–82.
- Chaboud, Alain P., Benjamin Chiquoine, Erik Hjalmarsson, and Clara Vega, 2014, Rise of the machines: Algorithmic trading in the foreign exchange market, *Journal of Finance* 69, 2045–2084.
- Chordia, Tarun, Avanidhar Subrahmanyam, and Qing Tong, 2014, Have capital market anomalies attenuated in the recent era of high liquidity and trading activity?, *Journal of Accounting and Economics* 58, 41–58.
- Cochrane, John H., 2005, *Asset Pricing* (Princeton University Press, Princeton, New Jersey).
- Cooper, Michael J., Huseyin Gulen, and Michael J. Schill, 2008, Asset growth and the cross-section of stock returns, *Journal of Finance* 63, 1609–1651.
- Datar, Vinay T., Narayan Y. Naik, and Robert Radcliffe, 1998, Liquidity and stock returns: An alternative test, *Journal of Financial Markets* 1, 203–219.
- De Long, J. Bradford, Andrei Shleifer, Lawrence H. Summers, and Robert J. Waldmann, 1990, Noise trader risk in financial markets, *Journal of Political Economy* 98, 703–738.
- Elton, Edwin J., 1999, Presidential address: Expected return, realized return, and asset pricing tests, *Journal of Finance* 54, 1199–1220.
- Fama, Eugene F., 1970, Efficient capital markets: A review of theory and empirical work, *Journal of Finance* 25, 383–417.
- Fama, Eugene F., 1991, Efficient capital markets: II, Journal of Finance 46, 1575–1617.
- Fama, Eugene F., and Kenneth R. French, 1992, The cross-section of expected stock returns, *Journal of Finance* 47, 427–465.
- Fama, Eugene F., and Kenneth R. French, 1993, Common risk factors in the returns on stocks and bonds, *Journal of Financial Economics* 33, 3–56.
- Fama, Eugene F., and Kenneth R. French, 2006, Profitability, investment and average returns, *Journal of Financial Economics* 82, 491–518.
- Fama, Eugene F., and James D. MacBeth, 1973, Risk, return, and equilibrium: Empirical tests, *Journal of Political Economy* 607–636.
- Friedman, Milton, 1953, The case for flexible exchange rates, in *Essays in Positive Economics* (University of Chicago Press, Chicago).

- Grinblatt, Mark, Matti Keloharju, and Juhani Linnainmaa, 2011, IQ and stock market participation, *Journal of Finance* 66, 2121–2164.
- Gromb, Denis, and Dimitri Vayanos, 2010, Limits of arbitrage: The state of the theory, Annual Review of Financial Economics 2, 251–275.
- Hanson, Samuel G., and Adi Sunderam, 2014, The growth and limits of arbitrage: Evidence from short interest, *Review of Financial Studies* 27, 1238–1286.
- Harvey, Campbell R., Yan Liu, and Heqing Zhu, 2015, ...and the crosssection of expected returns, *Review of Financial Studies*, forthcoming.
- Hausman, Jerry, 2001, Mismeasured variables in econometric analysis: Problems from the right and problems from the left, *Journal of Economic Perspectives* 15, 57–67.
- Hong, Harrison, Jeffrey D. Kubik, and Jeremy C. Stein, 2004, Social interaction and stock-market participation, *Journal of Finance* 59, 137–163.
- Hou, Kewei, and Mathijs van Dijk, 2012, Resurrecting the size effect: Firm size, profitability shocks, and expected stock returns, Ohio State University.
- Ince, Ozgur S., and R. Burt Porter, 2006, Individual equity return data from thomson datastream: Handle with care!, *Journal of Financial Research* 29, 463–479.
- Jegadeesh, Narasimhan, 1990, Evidence of predictable behavior of security returns, *Journal of Finance* 45, 881–898.
- Jegadeesh, Narasimhan, and Sheridan Titman, 1993, Returns to buying winners and selling losers: Implications for stock market efficiency, *Journal* of Finance 48, 65–91.
- Jones, Christopher S., and Lukasz Pomorski, 2013, Investing in disappearing anomalies, Working paper, University of Southern California.
- Kahneman, Daniel, and Amos Tversky, 1979, Prospect theory: An analysis of decision under risk, *Econometrica* 47, 263–291.
- Kumar, Alok, and Charles Lee, 2006, Retail investor sentiment and return comovements, *The Journal of Finance* 61, 2451–2486.
- Lintner, John, 1965, The valuation of risk assets and the selection of risky investments in stock portfolios and capital budgets, *Review of Economics* and Statistics 47, 13–37.

- Lo, Andrew W., 2004, The adaptive markets hypothesis: Market efficiency from an evolutionary perspective, *Journal of Portfolio Management, Forthcoming* 30, 15–29.
- Lowenstein, Roger, 2000, When Genius Failed: The Rise and Fall of Long-Term Capital Management (Random House, New York).
- Lucas, Robert E., Jr., 1978, Asset prices in an exchange economy, *Econo*metrica 46, 1429–1445.
- Mankiw, N. Gregory, and Stephen P. Zeldes, 1991, The consumption of stockholders and nonstockholders, *Journal of Financial Economics* 29, 97–112.
- McLean, R. David, and Jeffrey Pontiff, 2015, Does academic publication destroy predictability?, *Journal of Finance*, forthcoming.
- Merton, Robert C., 1973, An intertemporal capital asset pricing model, *Econometrica* 41, 867–887.
- Miller, Merton, and Myron Scholes, 1972, Rates of return in relation to risk: A reexamination of some recent findings, in Michael C. Jensen, ed., *Studies in the Theory of Capital Markets* (Praeger, New York).
- Nagel, Stefan, 2002, Is momentum caused by delayed overreaction?, Working paper, London Business School.
- Pontiff, Jeffrey, and Artemiza Woodgate, 2008, Share issuance and crosssectional returns, *Journal of Finance* 63, 921–945.
- Roll, Richard, 1994, What every CFO should know about scientific progress in financial economics: What is known and what remains to be resolved, *Financial Management* 69–75.
- Ross, Stephen A., 1976, The arbitrage theory of capital asset pricing, *Journal* of Economic Theory 13, 341–360.
- Schmidt, Peter S., Urs von Arx, Andreas Schrimpf, Alexander F. Wagner, and Andreas Ziegler, 2014, On the construction of common size, value and momentum factors in international stock markets: A guide with applications, Working paper, Swiss Finance Institute.
- Schwert, G. William, 2003, Anomalies and market efficiency, in George M. Constantinides, Milton Harris, and René M. Stulz, eds., Handbook of the Economics of Finance (Elsevier, Amsterdam).
- Shafir, Eldar, Peter Diamond, and Amos Tversky, 1997, Money illusion, Quarterly Journal of Economics 112, 341–374.

- Sharpe, William F., 1964, Capital asset prices: A theory of market equilibrium under conditions of risk, *Journal of Finance* 19, 425–442.
- Shiller, Robert J., 2000, *Irrational Exuberance* (Princeton University Press, Princeton, New Jersey).
- Shleifer, Andrei, 2000, Inefficient Markets: An Introduction to Behavioral Finance (Oxford University Press, Oxford).
- Shleifer, Andrei, and Robert W. Vishny, 1997, The limits of arbitrage, Journal of Finance 52, 35–55.
- Sloan, Richard G., 1996, Do stock prices fully reflect information in accruals and cash flows about future earnings?, *The Accounting Review* 71, 289– 315.
- Stambaugh, Robert F., 2014, Presidential address: Investment noise and trends, Journal of Finance 69, 1415–1453.
- Stein, Jeremy C., 2009, Presidential address: Sophisticated investors and market efficiency, *Journal of Finance* 64, 1517–1548.
- Subrahmanyam, Avanidhar, 2010, The cross-section of expected stock returns: What have we learnt from the past twenty-five years of research?, European Financial Management 16, 27–42.
- Sullivan, Ryan, Allan Timmermann, and Halbert White, 1999, Datasnooping, technical trading rule performance, and the bootstrap, *Journal* of Finance 54, 1647–1691.
- van Dijk, Mathijs A., 2011, Is size dead? a review of the size effect in equity returns, *Journal of Banking & Finance* 35, 3263–3274.
- van Rooij, Maarten, Annamaria Lusardi, and Rob Alessie, 2011, Financial literacy and stock market participation, *Journal of Financial Economics* 101, 449–472.

	definitions
	characteristic
ŕ	Firm
F	Lable 1:

Variable	Definition
ACC	Accounting accruals - the change in non-cash current assets minus the change in current liabilities all divided
	by total assets (Sloan, 1996)
AG	Asset growth - Yearly percentage change in total assets (Cooper et al., 2008)
$\rm B/M$	Book equity over market equity from previous December (Fama and French, 1992)
ISSUE	New issues - Change in number of shares outstanding from 11 months ago (Pontiff and Woodgate, 2008)
PROFIT	Profitability - Earnings over book equity (Fama and French, 2006)
$\mathbb{R}1$	Return lagged by 1 month (Jegadeesh, 1990)
R212	11 month cumulative return to the start of previous month (Jegadeesh and Titman, 1993)
SIZE	Market value of firm's equity (Banz, 1981)
TURN	Stock turnover - trading volume over number of shares outstanding (Datar et al., 1998)
This table pr	ovides definitions for the anomalous firm characteristic variables we analyse in this paper.

	Jan 1990 - Dec 2013	Jan 1990 - Dec 2001	Jan 2002 - Dec 2013
	Return $(\%)$	Return $(\%)$	Return $(\%)$
ACC	0.208	0.355	0.060
	(0.680)	(0.550)	(0.869)
AG	1.158^{**}	1.467^{*}	0.849
	(0.004)	(0.016)	(0.203)
$\rm B/M$	0.836^{*}	1.179^{**}	0.494
	(0.015)	(0.008)	(0.271)
ISSUE	-0.222	-0.124	-0.300
	(0.768)	(0.658)	(0.343)
PROFIT	1.907^{**}	2.110^{**}	1.704^{**}
	(0.000)	(0.00)	(0.00)
$\mathbf{R1}$	-0.209	-0.450	0.032
	(0.418)	(0.268)	(0.902)
R212	1.488^{**}	2.296^{**}	0.680
	(0.00)	(0.00)	(0.061)
SIZE	-0.112	0.070	-0.252
	(0.675)	(0.866)	(0.573)
TURN	1.857^{**}	2.237^{**}	1.4775^{**}
	(0.000)	(0.00)	(0.00)
The table show values of a give first column of first and seconc ** and * indic accruals measu divided by tot: the book to mi PROFIT is pro is the cumulati the market valu of shares outst	vs average monthly returns of an characteristic and short th results refers to the full sample l non-overlapping sub-sample ate significance at a 1% and red as the change in non-cas al assets. AG is asset growth arket ratio. ISSUE is the cha offtability measured as earnin, ve return in the 11 months p te of the firm's equity. TURN anding.	sorted portfolios long the quin e quintile with the lowest val le period. The second and thi 5% level, respectively. ACC h current assets minus the cl h measured as percentage ch age in number of shares outs gs over book equity. R1 is on rior to the previous month. ⁴ is stock turnover measured a	utile of stocks with the highest ues of the characteristic. The rd results columns refer to the intheses below each coefficient. J is the change in accounting ange in current liabilities all ange in total assets. B/M is itanding from 11 months ago. e-month lagged return. R212 SIZE is firm size measured as s trading volume over number

returns
portfolio
quintile
Long-short
Table 2:

	Jan 1990 - Dec 2013	Jan 1990 - Dec 2001	Jan 2002 - Dec 2013
	Return $(\%)$	Return $(\%)$	Return $(\%)$
ACC	0.266	0.182	0.661
	(0.517)	(0.697)	(0.344)
AG	0.964^{**}	1.483^{*}	0.445
	(0.001)	(0.020)	(0.309)
$\rm B/M$	0.629^{**}	0.782^{*}	0.477
	(0.003)	(0.022)	(0.096)
ISSUE	-0.707	-0.343	-1.071
	(0.263)	(0.644)	(0.120)
PROFIT	1.819^{**}	2.052^{**}	1.585^{**}
	(0.000)	(0.00)	(0.00)
$\mathbf{R1}$	-0.296	-0.810	0.218
	(0.444)	(0.076)	(0.681)
R212	1.473^{**}	2.069^{**}	0.877
	(0.00)	(0.00)	(0.502)
SIZE	0.191	0.050	0.332
	(0.494)	(0.895)	(0.421)
TURN	2.260^{**}	2.589^{**}	1.931^{**}
	(0.000)	(0.00)	(0.000)
The table shorvalues of a givv first column of first and seconc ** and * indic accruals measu divided by tot: the book to m: PROFIT is pro is the cumulati the market valu of shares outst	ws average monthly returns of en characteristic and short th results refers to the full sampl I non-overlapping sub-samples ate significance at a 1% and red as the change in non-cas al assets. AG is asset growth arket ratio. ISSUE is the cha offtability measured as earning we return in the 11 months p te of the firm's equity. TURN anding.	[±] sorted portfolios long the de ne decile with the lowest valu- le period. The second and thin s. P-values are stated in paren 5% level, respectively. ACC h current assets minus the cl n measured as percentage ch nge in number of shares outs gs over book equity. R1 is on rior to the previous month. ⁴ is stock turnover measured a	cile of stocks with the highest less of the characteristic. The rd results columns refer to the ntheses below each coefficient. \Im is the change in accounting hange in current liabilities all ange in total assets. B/M is standing from 11 months ago. te-month lagged return. R212 SIZE is firm size measured as s trading volume over number

Table 3: Long-short decile portfolio returns

29

	Jan 1990 -	- Dec 2013	Jan 1990 -	- Dec 2001	Jan 2002 -	Dec 2013
	Constant	FM Coef.	Constant	FM Coef.	Constant	FM Coef.
ACC	-0.001	0.043^{**}	0.001	0.054^{**}	0.003	0.027
	(0.420)	(0.00)	(0.432)	(0.00)	(0.385)	(0.065)
AG	-0.004	0.005^{*}	-0.003	0.009^{*}	0.000	0.002^{**}
	(0.190)	(0.013)	(0.279)	(0.022)	(0.483)	(0.008)
$\rm B/M$	-0.006	0.002^{**}	-0.003	0.002^{*}	-0.003	0.002
	(0.118)	(0.004)	(0.262)	(0.018)	(0.380)	(0.079)
ISSUE	-0.004	0.000	-0.002	0.000	-0.001	0.000
	(0.200)	(0.285)	(0.353)	(0.189)	(0.449)	(0.104)
PROFIT	-0.003	0.007^{**}	-0.002	0.010^{**}	0.001	0.004^{**}
	(0.246)	(0.00)	(0.335)	(0.001)	(0.451)	(0.000)
$\mathbf{R1}$	-0.004	-0.017^{**}	-0.001	-0.029^{**}	0.000	-0.007
	(0.229)	(0.005)	(0.392)	(0.002)	(0.479)	(0.162)
R212	0.000	0.011^{**}	-0.001	0.023^{**}	0.005	0.006
	(0.445)	(0.008)	(0.347)	(0.00)	(0.215)	(0.110)
SIZE	-0.017^{*}	0.001^{**}	-0.007	0.000	-0.018^{*}	0.001^{**}
	(0.024)	(0.005)	(0.294)	(0.271)	(0.041)	(0.000)
TURN	-0.012^{**}	0.009^{**}	-0.012^{**}	0.010^{**}	-0.007	0.008^{**}
	(0.004)	(0.000)	(0.003)	(0.000)	(0.195)	(0.000)
This table shows	univariate Fam	a-MacBeth regres	ssion coefficients	from risk-adjusted	d individual firm	returns on firm

Table 4: Univariate Fama-MacBeth regression coefficients

assets. AG is asset growth measured as percentage change in total assets. \ddot{B}/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as characteristics. We follow the procedure set out by Brennan et al. (1998) and risk-adjust the individual firm returns using the four-factor Fama and French (1993) and Carhart (1997) model. The table shows results for the full sample (January 1990 to December 2013), the first half of the sample (January 1990 to December 2001), and the second half of the sample (January 2002 to December 2013). Newey-West p-values are stated in parentheses below each coefficient. ** and * indicate significance at a 1% and 5% level, respectively. ACC is the change in accounting accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total | # trading volume over number of shares outstanding.

	ACC	AG	B/M	ISSUE	PROFIT	R1	R212	SIZE	TURN
ACC	1.000								
AG	0.066	1.000							
$\rm B/M$	0.068	0.080	1.000						
ISSUE	0.010	0.096	0.025	1.000					
PROFIT	0.070	0.031	-0.021	-0.009	1.000				
$\mathbf{R1}$	0.018	0.028	0.027	-0.004	0.055	1.000			
R212	-0.020	0.117	0.051	0.038	-0.029	0.199	1.000		
SIZE	0.018	-0.007	-0.245	0.061	0.101	0.018	-0.099	1.000	
TURN	0.015	-0.012	-0.093	-0.016	0.065	0.040	0.071	0.410	1.000
This table sh acteristics. Th	ows the time ie sample po	ie series ave eriod is Jan	trages of me uary 1990 t	o December	sectional corre 2013. ACC is	lations bet the change	ween the t in account	ransforme ing accrua	d firm char- ls measured
as the change	in non-cash	current ass	sets minus t	the change i	a current liabil	ities all div	vided by to	tal assets.	AG is at

	S
•	E
•	erist
-	characte
c	hrm
د	×
•	matrix
	Correlation
1	 C
Ē	Table

as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding.

			Full sa	mple: Jar	1 1990 - De	$c \ 2013$			
Constant	ACC	AG	$\rm B/M$	ISSUE	PROFIT	$\mathbb{R}1$	R212	SIZE	TURN
-0.002	0.036^{**}	0.006^{*}							
(0.342)	(0.002)	(0.012)							
-0.004	0.033^{**}	0.006^{**}	0.003^{**}						
(0.186)	(0.006)	(0.006)	(0.001)						
-0.004	0.034^{**}	0.007^{**}	0.002^{**}	0.000					
(0.199)	(0.005)	(0.003)	(0.001)	(0.057)					
-0.005	0.020^{*}	0.006^{**}	0.003^{**}	0.000	0.008^{**}				
(0.157)	(0.041)	(0.002)	(0.000)	(0.124)	(0.000)				
-0.005	0.021^{*}	0.006^{**}	0.003^{**}	0.000	0.008^{**}	-0.001			
(0.173)	(0.032)	(0.001)	(0.000)	(0.138)	(0.000)	(0.427)			
-0.004	-0.002	0.004^{**}	0.003^{**}	0.000^{**}	0.018^{**}	-0.009	0.013^{**}		
(0.125)	(0.452)	(0.005)	(0.003)	(0.002)	(0.000)	(060.0)	(0.002)		
0.005	-0.001	0.004^{**}	0.002^{*}	0.000^{**}	0.019^{**}	-0.011	0.013^{**}	-0.001^{*}	
(0.226)	(0.455)	(0.006)	(0.015)	(0.008)	(0.000)	(0.066)	(0.002)	(0.032)	
-0.015^{**}	0.003	0.004^{**}	0.007^{**}	0.000	0.024^{**}	-0.027**	0.006		0.009^{**}
(0.000)	(0.420)	(0.00)	(0.000)	(0.066)	(0.003)	(0.002)	(0.065)		(0.000)
0.033^{**}	0.011	0.003	0.005^{**}	0.000	0.024^{**}	-0.038**	0.005	-0.004^{**}	0.011^{**}
(0.001)	(0.238)	(0.061)	(0.000)	(0.124)	(0.001)	(0.00)	(0.137)	(0.000)	(0.000)
This table she	ows Fama-M	acBeth regre	ssion coefficie	ents from fire	m returns on fi	irm characteri	istics. We fol	low the proce	dure set out
by Brennan et	; al. (1998) ;	and risk-adju	ist the indivi-	dual firm ret	turns using th	e four-factor	Fama and Fr	rench (1993) a	and Carhart
(1997) model.	The sample	period is Ja	vnuary 1990 t	to December	2013. Newey-	West p-value	s are stated i	n parentheses	s below each

Table 6: Multivariate Fama-MacBeth regression coefficients

coefficient. ** and * indicate significance at ă 1% and 5% level, respectively. ĂCC is the change in accounting accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding.

			Sub-sai	nple I: Ja	n 1990 - De	sc 2001			
onstant	ACC	AG	$\rm B/M$	ISSUE	PROFIT	$\mathbf{R1}$	R212	SIZE	TURN
-0.001	0.045^{**}	0.010^{*}							
(0.421)	(0.004)	(0.022)							
-0.003	0.038^{*}	0.011^{*}	0.002^{*}						
(0.285)	(0.021)	(0.011)	(0.013)						
-0.003	0.038^{*}	0.011^{**}	0.002^{*}	0.000					
(0.288)	(0.023)	(0.007)	(0.015)	(0.439)					
-0.004	0.023	0.010^{**}	0.003^{**}	0.000	0.010^{**}				
(0.200)	(0.096)	(0.004)	(0.003)	(0.431)	(0.00)				
-0.004	0.024	0.010^{**}	0.002^{**}	0.000	0.010^{**}	-0.014			
(0.215)	(0.073)	(0.003)	(0.004)	(0.392)	(0.000)	(0.110)			
0.008^{*}	-0.006	0.007^{**}	0.003^{*}	0.000^{*}	0.021^{**}	-0.004	0.023^{**}		
(0.030)	(0.370)	(0.002)	(0.020)	(0.036)	(0.000)	(0.360)	(0.000)		
0.005	-0.005	0.007^{**}	0.002	0.000	0.022^{**}	-0.007	0.023^{**}	-0.001	
0.316)	(0.380)	(0.002)	(0.102)	(0.077)	(0.000)	(0.256)	(0.000)	(0.052)	
0.021^{**}	0.009	0.007^{*}	0.007^{**}	0.000	0.028^{*}	-0.042^{**}	0.017^{**}		0.008^{**}
(0.000)	(0.355)	(0.015)	(0.000)	(0.168)	(0.013)	(0.002)	(0.005)		(0.000)
0.024	0.017	0.005	0.006^{**}	0.000	0.031^{**}	-0.049^{**}	0.019^{**}	-0.004^{**}	0.010^{**}
(0.069)	(0.266)	(0.062)	(0.005)	(0.238)	(0.010)	(0.001)	(0.005)	(0.000)	(0.000)
nis table shc Brennan et	ws Fama-Mi al. (1998) a	acBeth regre und risk-adju	ssion coefficients st the individ	ents from fir dual firm ret	m returns on fi turns using the	rm character • four-factor	istics. We fol Fama and Fr	low the proce rench (1993)	dure set out and Carhart
	mi o como lo	- I	1000 +		N North	TATE	atotod	·	

Table 7: Multivariate Fama-MacBeth regression coefficients

the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding. (1997) model. The sample period is January 1990 to December 2001. Newey-West p-values are stated in parentheses below each coefficient. ** and * indicate significance at a 1% and 5% level, respectively. ACC is the change in accounting accruals measured as

			Sub-san	nple II: Ja	ın 2002 - D	$ec \ 2013$			
Constant	ACC	AG	$\rm B/M$	ISSUE	PROFIT	$\mathbf{R1}$	R212	SIZE	TURN
0.002	0.020	0.003^{*}							
(0.416)	(0.118)	(0.010)							
-0.001	0.020	0.002^{*}	0.002^{*}						
(0.467)	(0.117)	(0.013)	(0.020)						
0.000	0.021	0.003^{**}	0.002^{*}	0.000^{*}					
(0.479)	(0.104)	(0.005)	(0.034)	(0.036)					
0.000	0.009	0.003^{**}	0.002^{*}	0.000	0.006^{**}				
(0.477)	(0.275)	(0.006)	(0.029)	(0.085)	(0.000)				
0.000	0.010	0.003^{**}	0.002^{*}	0.000	0.006^{**}	-0.014^{**}			
(0.482)	(0.255)	(0.005)	(0.030)	(0.078)	(0.000)	(0.008)			
0.002	-0.009	0.001	0.002	0.000^{*}	0.015^{**}	-0.007	0.006		
(0.390)	(0.274)	(0.165)	(0.081)	(0.028)	(0.000)	(0.228)	(0.080)		
0.008	-0.009	0.001	0.002	0.000^{*}	0.016^{**}	-0.007	0.006	0.000	
(0.191)	(0.281)	(0.193)	(0.054)	(0.036)	(0.000)	(0.237)	(0.104)	(0.154)	
-0.006	-0.009	0.002	0.005^{**}	0.000	0.013^{**}	-0.013	0.002		0.008^{**}
(0.140)	(0.254)	(0.151)	(0.001)	(0.223)	(0.006)	(0.081)	(0.295)		(0.000)
0.046^{**}	-0.006	0.001	0.003^{*}	0.000	0.016^{**}	-0.019^{*}	-0.003	-0.004**	0.011^{**}
(0.000)	(0.340)	(0.188)	(0.019)	(0.273)	(0.003)	(0.029)	(0.156)	(0.000)	(0.000)
This table sho by Brennan et	ws Fama-Mi al. (1998) a	acBeth regre und risk-adju	ssion coefficie st the indivi	ents from firn dual firm ret	m returns on fi	rm character e four-factor	istics. We fol Fama and Fi	low the proce rench (1993)	edure set out and Carhart
(1997) model. coefficient. ** ε	'I'he sample and * indicat	period is Ja te significanc	nuary 2002 t e at a 1% and	o December d 5% level. re	2013. Newey- sspectivelv. AC	West p-values CC is the char	s are stated i 1ge in accoun	in parenthese iting accruals	s below each measured as
the change in n	on-cash curi	rent assets m	inus the chan	ge in current	liabilities all c	livided by tot	al assets. AG	is asset grow	th measured
as percentage c 11 months ago	change in to PROFIT	tal assets. B, is profitabili	/M is the boo tv measured	ok to market as earnings	over book equ	is the change uity. R1 is o	in number o ne-month las	t shares outst gged return.	anding from R212 is the
cumulative ret	urn in the 1	1 months pr	ior to the pr	evious mont	h. SIZE is fir	m size measu	red as the n	narket value	of the firm's
equity. TURN	is stock turi	nover measu	red as trading	g volume ove	er number of sh	nares outstan	ding.		

Table 8: Multivariate Fama-MacBeth regression coefficients

TURN	0.068^{**}	4.140
SIZE	-0.046^{**}	-3.931
R212	-2.092^{**}	-32.142
$\mathbf{R1}$	3.242^{**}	15.752
PROFIT	-0.635^{**}	-3.970
ISSUE	-0.000**	-3.124
$\rm B/M$	-0.095**	-5.109
AG	-0.0778**	-2.605
ACC	0.475	1.430
Variable	Trend coef.	t-stat

Table 9: Time trend t-statistics of Fama-MacBeth coefficients

accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding. This table provides the time trend t-statistic from regressions of 60-month moving average Fama-MacBeth multiple regression (returns on all nine anomaly variables) coefficients on time. The we adjust for risk in the regressions using the four-factor Fama and French (1993) and Carhart (1997) model. ** and * indicate significance at a 1% and 5% level, respectively. ACC is the change in accounting

Figure 1: Interpreting an anomaly



This figure illustrates the process of investigating an anomaly's economic significance. First, the anomaly may be a spurious result produced by data mining. Second, a nonspurious anomaly may compensate investors for bearing risk or it may indicate informational inefficiency. Third, an inefficiency may be exploitable and could generate abnormal returns or it could be unexploitable due to trading costs and more complex restrictions on investor behaviour.



Figure 2: 60-month moving averages of simple Fama-MacBeth coefficients

This figure shows 60-month moving averages of Fama-MacBeth coefficients from multiple regressions of risk-adjusted returns on all nine firm characteristic over the full sample period using the Fama and French (1993) and Carhart (1997) factors to adjust for risk. ACC is the change in accounting accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding.

Appendix

A.1 Alternative Fama-MacBeth risk-adjustment

In this section, we re-examine the significance of our nine anomaly variables using alternative methods of risk-adjustment for individual firm returns. In the main body of the paper, we risk-adjust individual firm returns using the four-factor Fama and French (1993) and Carhart (1997) model. Here we repeat the univariate and multivariate Fama-MacBeth regressions from Tables 4, 6, 7, and 8 using the three-factor Fama and French (1993) model and the CAPM.

A.2 Univariate Fama-MacBeth regressions

We first repeat our univariate Fama-MacBeth analysis from Section 6.2 using the three-factor Fama and French (1993) model and the one-factor CAPM to adjust for risk.

Table 10 presents these results based on returns adjusted using the threefactor Fama and French (1993) model. Table 11 presents these same results, but using the one-factor CAPM to adjust firm returns for risk. In each case, risk-adjusted returns are regressed on a constant and one specified anomaly variable. Thus, each regression produces a constant coefficient and one anomaly variable coefficient.

We follow the approach of our early work and analyse the data over three time periods. The full sample period of January 1990 to December 2013. The first sub-sample runs from January 1990 to December 2001. The second sub-sample runs from January 2002 to December 2013.

<Table 10>

<Table 11>

Our results are remarkably similar to those in Table 4. We see that omitting the Carhart (1997) "momentum" factor has little impact on our results and we draw the same conclusions regarding anomaly attenuation. The first column of Table 10 and Table 11 both refer to the results for the full sample period. We see that univariate Fama-MacBeth coefficients are strongly significant for the full sample period. In both cases, seven of the nine coefficients are significant at a 1% level and eight of the coefficients are significant at a 5% level. The second and third columns of Tables 10 and 11 show our results for the first and second sub-samples, respectively. We find that the significance of these anomaly coefficients is most pronounced in the first sub-sample. The signs of these coefficients also correspond closely to our earlier results. Similarly to Table 4, we find that the asset growth, profitability, size, and turnover coefficients remain significant in this more recent time period.

A.2.2 Multivariate Fama-MacBeth regressions

Finally, we repeat our multivariate Fama-MacBeth regression analysis using the three-factor Fama and French (1993) model and the one-factor CAPM. Our results are again very similar to those in the main body of the paper.

Tables 12, 13, and 14 present Fama-MacBeth results for which individual firms are adjusted using the three-factor Fama and French (1993) model. They correspond to Tables 6, 7, and 8 of the main body of the paper. Tables 15, 16, and 17 present Fama-MacBeth results for which individual firms are adjusted using the one-factor CAPM. They again correspond to Tables 6, 7, and 8 of the main body of the paper.

We find that the profitability and turnover effects are the most robust of our set of anomalies to specification changes and the use of alternate subsamples. This is consistent with our earlier findings. We find less reliable significance for the book-to-market and size effects.

Interestingly, we find that momentum effect is insignificant in our second sub-sample for all three models of expected returns we consider. We also find that the accruals and asset growth anomalies are quite sensitive to our choice of regression specification.

In summary, our Fama-MacBeth regression findings are robust to the use of the four-factor Fama and French (1993) and Carhart (1997) model, the three-factor Fama and French (1993) model, and the one-factor CAPM.

<Table 12> <Table 13> <Table 14> <Table 15> <Table 16> <Table 17>

	Jan 1990 -	- Dec 2013	Jan 1990 -	- Dec 2001	Jan 2002 -	- Dec 2013
	Constant	FM Coef.	Constant	FM Coef.	Constant	FM Coef.
ACC	-0.001	0.043^{**}	0.000	0.055^{**}	0.000	0.027
	(0.400)	(0.00)	(0.470)	(0.00)	(0.479)	(0.065)
AG	-0.005	0.005^{*}	-0.003	0.008^{*}	-0.003	0.002^{*}
	(0.179)	(0.012)	(0.259)	(0.024)	(0.348)	(0.014)
$\rm B/M$	-0.006	0.002^{**}	-0.004	0.003^{**}	-0.006	0.002
	(0.107)	(0.003)	(0.219)	(0.005)	(0.253)	(0.074)
ISSUE	-0.005	0.000	-0.002	0.000	-0.004	0.000
	(0.186)	(0.236)	(0.325)	(0.209)	(0.317)	(0.055)
PROFIT	-0.004	0.008^{**}	-0.003	0.010^{**}	-0.002	0.004^{**}
	(0.234)	(0.00)	(0.302)	(0.001)	(0.404)	(0.000)
$\mathbf{R1}$	-0.004	0.020^{**}	-0.002	0.031^{**}	-0.004	0.009
	(0.218)	(0.001)	(0.370)	(0.001)	(0.342)	(0.099)
R212	0.000	0.011^{**}	-0.001	0.021^{**}	0.003	(0.004)
	(0.460)	(0.009)	(0.353)	(0.000)	(0.342)	(0.241)
SIZE	-0.017^{*}	0.001^{**}	-0.006	0.000	-0.024^{*}	0.002^{**}
	(0.023)	(0.005)	(0.325)	(0.340)	(0.015)	(0.000)
TURN	-0.013^{**}	0.009^{**}	-0.012^{**}	0.009^{**}	-0.010	0.008^{**}
	(0.003)	(0.000)	(0.003)	(0.000)	(0.110)	(0.000)
This table shows characteristics. W	s univariate Fam Ve follow the pro	a-MacBeth regres	ssion coefficients Z Brennan et al.	from risk-adjusted (1998) and risk-ad	d individual firm diust the individu	returns on firm ual firm returns

Table 10: Univariate Fama-MacBeth regression coefficients

using the three-factor Fama and French (1993) model. The table shows results for the full sample (January 1990 to December 2013), the first half of the sample (January 1990 to December 2011), and the second half of the sample * indicate significance at a 1% and 5% level, respectively. ACC is the change in accounting accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in \mathbf{ns} number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE | # (January 2002 to December 2013). Newey-West p-values are stated in parentheses below each coefficient. ** and is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding.

)		
	Jan 1990.	- Dec 2013	Jan 1990 .	- $Dec 2001$	Jan 2002 .	- Dec 2013
	Constant	FM Coef.	Constant	FM Coef.	Constant	FM Coef.
ACC	-0.001	0.044^{**}	0.000	0.060^{**}	-0.001	0.030
	(0.440)	(0.00)	(0.488)	(0.00)	(0.466)	(0.059)
AG	-0.004	0.005^{*}	-0.004	0.009^{*}	-0.004	0.002^{*}
	(0.207)	(0.012)	(0.222)	(0.022)	(0.329)	(0.015)
$\rm B/M$	-0.006	0.003^{**}	-0.005	0.003^{**}	-0.006	0.002^{*}
	(0.127)	(0.003)	(0.182)	(0.005)	(0.232)	(0.044)
ISSUE	-0.004	0.000	-0.003	0.000	-0.005	0.000
	(0.216)	(0.287)	(0.285)	(0.199)	(0.300)	(0.058)
PROFIT	-0.003	0.008^{**}	-0.003	0.011^{**}	-0.003	0.004^{**}
	(0.266)	(0.000)	(0.265)	(0.000)	(0.386)	(0.00)
$\mathbf{R1}$	-0.004	0.024^{**}	-0.002	0.035^{**}	-0.004	0.012
	(0.248)	(0.00)	(0.329)	(0.00)	(0.329)	(0.065)
R212	0.001	0.012^{**}	-0.001	0.020^{**}	0.003	0.004
	(0.404)	(0.007)	(0.353)	(0.001)	(0.333)	(0.236)
SIZE	-0.017^{*}	0.001^{**}	-0.006	0.000	-0.024^{*}	0.002^{**}
	(0.030)	(0.006)	(0.324)	(0.363)	(0.017)	(0.00)
TURN	-0.012^{**}	0.009^{**}	-0.014^{**}	0.010^{**}	-0.011	0.008^{**}
	(0.005)	(0.000)	(0.002)	(0.000)	(0.104)	(0.000)
This table show:	s univariate Fam	a-MacBeth regre	ssion coefficients	from risk-adjuste	d individual firm	returns on firm
using the one-fac	tor CAPM. The	sceaure set out p table shows rest	y <u>brennan et al.</u> ilts for the full se	(1990) and risk-a mple (January 19	ajust the maivia 990 to December	. 2013), the first
half of the sampl	e (January 1990	to December 200	01), and the seco	nd half of the san	nple (January 20	02 to December
2013). Newey-W	est p-values are	stated in parenth	teses below each	coefficient. ** and	d * indicate signi	ificance at a 1%
and 5% level, re-	spectively. ACC	is the change in	accounting accr	uals measured as	the change in n	on-cash current
assets minus the change in total a	cnange in currei ssets. B/M is th	te book to market	ratio. ISSUE is	the change in nu	growth measure mber of shares ou	ed as percentage utstanding from
11 months ago. F	ROFIT is profit	ability measured	as earnings over l	oook equity. R1 is	s one-month lagg	ed return. R212
is the cumulative value of the firm	e return in the 1 's equity. TURN	1 months prior t is stock turnove	to the previous n r measured as tra	nonth. SIZE is fi ading volume ove:	rm size measured r number of shar	d as the market es outstanding.

Table 11: Univariate Fama-MacBeth regression coefficients

			Full sa	mple: Jan	1 1990 - De	c 2013			
Constant	ACC	AG	B/M	ISSUE	PROFIT	R1	R212	SIZE	TURN
-0.002	0.035^{**}	0.006^{*}							
(0.325)	(0.004)	(0.013)							
-0.005	0.032^{**}	0.006^{**}	0.003^{**}						
(0.169)	(0.009)	(0.007)	(0.000)						
-0.005	0.033^{**}	0.007^{**}	0.003^{**}	0.000					
(0.179)	(0.008)	(0.003)	(0.001)	(0.057)					
-0.005	0.018	0.006^{**}	0.003^{**}	0.000	0.008^{**}				
(0.139)	(0.066)	(0.002)	(0.000)	(0.127)	(0.000)				
-0.005	0.018	0.006^{**}	0.003^{**}	0.000	0.008^{**}	0.000			
(0.155)	(0.054)	(0.002)	(0.000)	(0.146)	(0.000)	(0.478)			
-0.005	-0.003	0.004^{**}	0.003^{**}	0.000^{**}	0.019^{**}	-0.008	0.012^{**}		
(0.108)	(0.397)	(0.006)	(0.002)	(0.001)	(0.000)	(0.120)	(0.002)		
0.004	-0.003	0.004^{**}	0.002^{*}	0.000^{**}	0.020^{**}	-0.009	0.012^{**}	-0.001^{*}	
(0.270)	(0.391)	(0.007)	(0.009)	(0.005)	(0.000)	(0.092)	(0.002)	(0.042)	
-0.015^{**}	0.003	0.004^{*}	0.007^{**}	0.000^{*}	0.021^{**}	-0.029**	0.006		0.008^{**}
(0.000)	(0.422)	(0.015)	(0.000)	(0.038)	(0.000)	(0.001)	(0.075)		(0.000)
0.032^{**}	0.009	0.003	0.005^{**}	0.000	0.024^{**}	-0.036**	0.005	-0.004**	0.011^{**}
(0.001)	(0.289)	(0.066)	(0.000)	(0.116)	(0.000)	(0.00)	(0.148)	(0.000)	(0.000)
This table sho by Brennan et	ws Fama-Mi al. (1998) a	acBeth regree ind risk-adju	st the individ	ants from firr dual firm ret	n returns on fi urns using the	rm characteri three-factor	Extrema and F	low the proce rench (1993)	dure set out model. The

Table 12: Multivariate Fama-MacBeth regression coefficients

sample period is January 1990 to December 2013. Newey-West p-values are stated in parentheses below each coefficient. ** and * indicate significance at a 1% and 5% level, respectively. ACC is the change in accounting accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding volume section.

			Sub-sar	nple I: Ja	n 1990 - De	sc 2001			
Constant	ACC	AG	B/M	ISSUE	PROFIT	R1	R212	SIZE	TURN
-0.001	0.045^{**}	0.009^{*}							
(0.399)	(0.005)	(0.028)							
-0.004	0.038^{*}	0.010^{*}	0.003^{**}						
(0.241)	(0.027)	(0.015)	(0.004)						
-0.004	0.036^{*}	0.010^{**}	0.003^{**}	0.000					
(0.244)	(0.031)	(600.0)	(0.004)	(0.449)					
-0.005	0.021^{*}	0.009^{**}	0.003^{**}	0.000	0.010^{**}				
(0.163)	(0.120)	(0.006)	(0.001)	(0.443)	(0.000)				
-0.005	0.021	0.009^{**}	0.003^{**}	0.000	0.010^{**}	0.015			
(0.186)	(0.095)	(0.005)	(0.001)	(0.383)	(0.000)	(0.087)			
-0.008*	-0.007	0.007^{**}	0.003^{**}	0.000^{*}	0.022^{**}	-0.003	0.021^{**}		
(0.021)	(0.335)	(0.004)	(0.006)	(0.044)	(0.000)	(0.392)	(0.000)		
0.005	-0.007	0.006^{**}	0.003^{*}	0.000	0.023^{**}	-0.006	0.022^{**}	-0.001	
(0.345)	(0.333)	(0.004)	(0.047)	(0.094)	(0.000)	(0.282)	(0.000)	(0.061)	
-0.021^{**}	0.004	0.006^{*}	0.008^{**}	0.000	0.029^{**}	-0.040^{**}	0.016^{*}		0.008^{**}
(0.000)	(0.426)	(0.024)	(0.000)	(0.186)	(0.006)	(0.004)	(0.011)		(0.00)
0.022	0.011	0.005	0.007^{**}	0.000	0.032^{**}	-0.048^{**}	0.018^{*}	-0.004^{**}	0.010^{**}
(0.075)	(0.332)	(0.086)	(0.001)	(0.261)	(0.006)	(0.001)	(0.011)	(0.001)	(0.00)
This table she by Brennan et	ows Fama-Mi al. (1998) a	acBeth regre nd risk-adju	ssion coefficie st the individ	ents from firn lual firm ret	m returns on fi turns using the	rm characteri three-factor	istics. We fol Fama and F	low the proce rench (1993)	dure set out model. The

Table 13: Multivariate Fama-MacBeth regression coefficients

sample period is January 1990 to December 2001. Newey-West p-values are stated in parentheses below each coefficient. ** and * indicate significance at a 1% and 5% level, respectively. ACC is the change in accounting accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding.

	ACC 0.021	AG 0.003**	Sub-sam B/M	ple II: Ja ISSUE	n 2002 - Do PROFIT	ec 2013 R1	R212	SIZE	TURN
(0.108) (0.021	<u> </u>	(0.015) (0.002^{*})	0.002^{*}						
(0.103) (0.023* ((0.019) 0.003^{**}	(0.019) 0.002^{*}	0.000^{*}					
(0.088)		(0.007)	(0.033)	(0.024)	**2000				
(0.245)		(800.0)	(0.029)	(0.054)	(0.001)				
0.012		0.003^{**}	0.002^{*}	0.000^{*}	0.006^{**}	-0.012^{*}			
(0.225)		(0.007)	(0.030)	(0.049)	(0.001)	(0.016)			
-0.003		0.001	0.002	0.000^{*}	0.016^{**}	-0.006	0.005		
(0.415)		(0.224)	(0.105)	(0.017)	(0.000)	(0.254)	(0.171)		
-0.004		0.001	0.002	0.000^{*}	0.017^{**}	-0.005	0.004	0.000	
(0.413)		(0.252)	(0.068)	(0.024)	(0.000)	(0.271)	(0.205)	(0.190)	
-0.003		0.001	0.004^{**}	0.000	0.014^{**}	-0.012	0.000		0.008^{**}
(0.422)		(0.215)	(0.001)	(0.148)	(0.003)	(0.090)	(0.455)		(0.000)
0.000		0.001	0.003^{*}	0.000	0.017^{**}	-0.018^{*}	-0.005	-0.004^{**}	0.011^{**}
(0.497)		(0.261)	(0.025)	(0.206)	(0.002)	(0.033)	(0.101)	(0.00)	(0.000)
ws Fama-Mac e et al. (1998) iod is Januar; iignificance at rent assets m PROFIT is rn in the 11	1 is as at at	Beth regre and risk-s y 2002 to I a 1% and inus the cl inus the cl sets. B/M profitabilit months pri	djust the in- djust the in- becember 207 5% level, res nange in cur is the book y measured or to the pr	ents from fin dividual firm 13. Newey-W pectively. A rent liabiliti to market r as earnings evious mont!	im returns on a returns using dest p-values <i>z</i> CC is the cha. CC is the cha. es all divided atio. ISSUE is over book equ h. SIZE is firr	firm charact g the three-fi are stated in nge in accoun by total ass s the change inty. R1 is o m size meast	ceristics. We actor Fama a parentheses tring accrua tets. AG is a in number o ne-month la ured as the 1	e follow the r and French (below each c ls measured a asset growth of shares outsi gged return. market value	rocedure set (993) model. sefficient. ** s the change measured as anding from R212 is the of the firm's
s stock turnover measured as trac	nover measured as trac	ed as trac	jing	g volume ove	r number of s	hares outstar	ıding.		

Table 14: Multivariate Fama-MacBeth regression coefficients

Constant	ACC	AG	B/M	ISSUE	PROFIT	R1	R212	SIZE	TURN
-0.002	0.038^{**}	0.006^{*}							
(0.365)	(0.003)	(0.012)							
-0.004	0.034^{**}	0.006^{**}	0.003^{**}						
(0.197)	(0.007)	(0.008)	(0.000)						
-0.004	0.034^{**}	0.007^{**}	0.003^{**}	0.000					
(0.208)	(0.007)	(0.003)	(0.003)	(0.082)					
-0.005	0.020	0.006^{**}	0.003^{**}	0.000	0.008^{**}				
(0.163)	(0.056)	(0.002)	(0.000)	(0.161)	(0.000)				
-0.005	0.020^{*}	0.006^{**}	0.003^{**}	0.000	0.008^{**}	0.004			
(0.179)	(0.046)	(0.002)	(0.000)	(0.182)	(0.000)	(0.274)			
-0.004	-0.004	0.004^{**}	0.003^{**}	0.000^{**}	0.019^{**}	-0.004	0.012^{**}		
(0.132)	(0.384)	(0.007)	(0.002)	(0.001)	(0.000)	(0.272)	(0.001)		
0.005	-0.004	0.004^{**}	0.002^{*}	0.000^{**}	0.020^{**}	-0.005	0.012^{**}	-0.001^{*}	
(0.236)	(0.382)	(0.009)	(0.009)	(0.006)	(0.000)	(0.209)	(0.001)	(0.035)	
-0.015^{**}	0.000	0.004^{**}	0.007^{**}	0.000	0.022^{**}	-0.025^{**}	0.005		0.009^{**}
(0.000)	(0.485)	(0.009)	(0.000)	(0.058)	(0.000)	(0.003)	(0.109)		(0.000)
0.033^{**}	0.005	0.003^{*}	0.005^{**}	0.000	0.025^{**}	-0.032^{**}	0.004	-0.004**	0.011^{**}
(0.001)	(0.352)	(0.050)	(0.000)	(0.131)	(0.001)	(0.000)	(0.173)	(0.000)	(0.000)
This table she sample period	is January 1	acBeth regre 1990 to Dece	ssion coeffici- mber 2013.	ents from C ^A Newev-West	APM-adjusted p-values are st	individual fir tated in pare	m returns on ntheses belov	. firm charact v each coeffic	eristics. The ient. ** and

Table 15: Multivariate Fama-MacBeth regression coefficients

* indicate significance at a 1% and 5% level, respectively. ACC is the change in accounting accruals measured as the change in non-cash current assets minus the change in current liabilities all divided by total assets. AG is asset growth measured as percentage change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding.

	0	-	Ins-unc	inple I: Ja	<u>II 1990 - D(</u>				
Constant	ACC	AG	$\rm B/M$	ISSUE	PROFIT	$\mathbf{R1}$	R212	SIZE	TURN
-0.002	0.050^{**}	0.009^{*}							
(0.361)	(0.005)	(0.027)							
-0.004	0.042^{*}	0.010^{*}	0.003^{**}						
(0.202)	(0.026)	(0.015)	(0.003)						
-0.004	0.040^{*}	0.010^{**}	0.003^{**}	0.000					
(0.205)	(0.030)	(0.00)	(0.003)	(0.385)					
-0.006	0.023	0.009^{**}	0.003^{**}	0.000	0.011^{**}				
(0.132)	(0.114)	(0.006)	(0.001)	(0.380)	(0.000)				
-0.006	0.022	0.009^{**}	0.003^{**}	0.000	0.011^{**}	0.019^{*}			
(0.151)	(0.095)	(0.005)	(0.000)	(0.315)	(0.000)	(0.042)			
-0.009*	-0.010	0.007^{**}	0.004^{**}	0.000^{*}	0.023^{**}	0.002	0.020^{**}		
(0.016)	(0.259)	(0.003)	(0.003)	(0.047)	(0.000)	(0.418)	(0.000)		
0.005	-0.010	0.007^{**}	0.003^{*}	0.000	0.024^{**}	-0.001	0.021^{**}	-0.001	
(0.349)	(0.249)	(0.003)	(0.042)	(0.103)	(0.000)	(0.444)	(0.000)	(0.057)	
-0.023^{**}	0.000	0.007^{**}	0.009^{**}	0.000	0.031^{**}	-0.036^{*}	0.012^{*}		0.008^{**}
(0.000)	(0.497)	(600.0)	(0.000)	(0.191)	(0.005)	(0.012)	(0.045)		(0.000)
0.021	0.007	0.005	0.007^{**}	0.000	0.034^{**}	-0.044**	0.014^{*}	-0.004**	0.010^{**}
(0.091)	(0.380)	(0.057)	(0.001)	(0.261)	(0.006)	(0.003)	(0.024)	(0.001)	(0.000)
This table she	ows Fama-Ma	acBeth regre	ssion coefficie	ents from C ⁴	APM-adjusted	individual fir	m returns on	firm charact	eristics. The
sample period	is January 1	100 to Dece	mber 2001.	Newey-West	p-values are s	tated in pare	ntheses belov	v each coeffic	ient. ** and
non-cash curre	inicance at a nt assets min	us the chang	e in current l	iabilities all d	divided by tota	all accountures all assets. AG	ig accruais it is asset growt	heasureu as u th measured a	ue cuange m us percentage

change in total assets. B/M is the book to market ratio. ISSUE is the change in number of shares outstanding from 11 months ago. PROFIT is profitability measured as earnings over book equity. R1 is one-month lagged return. R212 is the cumulative return in the 11 months prior to the previous month. SIZE is firm size measured as the market value of the firm's equity. TURN is stock turnover measured as trading volume over number of shares outstanding.

Table 16: Multivariate Fama-MacBeth regression coefficients

			Sub-sam	ıple II: Ja	n 2002 - De	ec 2013			
Constant	ACC	AG	B/M	ISSUE	PROFIT	R1	R212	SIZE	TURN
-0.001	0.023	0.003^{*}							
(0.436)	(0.095)	(0.019)							
-0.004	0.024	0.002^{*}	0.003^{**}						
(0.319)	(0.089)	(0.020)	(0.009)						
-0.004	0.026	0.003^{**}	0.003^{*}	0.000^{*}					
(0.330)	(0.076)	(0.007)	(0.014)	(0.025)					
-0.004	0.013	0.003^{**}	0.003^{*}	0.000	0.006^{**}				
(0.326)	(0.210)	(0.008)	(0.011)	(0.052)	(0.001)				
-0.004	0.014	0.003^{**}	0.003^{*}	0.000^{*}	0.006^{**}	-0.010^{*}			
(0.337)	(0.192)	(0.007)	(0.013)	(0.047)	(0.001)	(0.041)			
0.000	0.000	0.001	0.002^{*}	0.000^{*}	0.016^{**}	-0.008	0.005		
(0.497)	(0.500)	(0.187)	(0.049)	(0.026)	(0.000)	(0.175)	(0.144)		
0.006	0.000	0.001	0.002^{*}	0.000^{*}	0.017^{**}	-0.008	0.005	0.000	
(0.233)	(0.499)	(0.211)	(0.037)	(0.034)	(0.000)	(0.192)	(0.176)	(0.145)	
-0.008	0.000	0.001	0.005^{**}	0.000	0.014^{**}	-0.015^{*}	0.001		0.009^{**}
(0.076)	(0.499)	(0.192)	(0.000)	(0.186)	(0.004)	(0.044)	(0.423)		(0.000)
0.045^{**}	0.003	0.001	0.003^{*}	0.000	0.017^{**}	-0.021^{*}	-0.005	-0.005**	0.012^{**}
(0.000)	(0.429)	(0.226)	(0.014)	(0.246)	(0.002)	(0.014)	(0.101)	(0.000)	(0.000)
This table sho	ws Fama-Ma	acBeth regres	sion coefficie	ints from CA	PM-adjusted i	individual fir	m returns on	firm charact	eristics. The
sample period	is January 2	002 to Decer	nber 2013. ľ	Newey-West	p-values are st	ated in pare	ntheses belov	w each coeffic	ient. ** and
in non-cash cui	rrent assets	minus the ch	ange in cur	rent liabiliti	es all divided	by total ass	ung accruais ets. AG is a	s measureu a: asset growth	measured as
percentage chai	nge in total	assets. B/M	is the book	to market r	atio. ISSUE is	the change	in number o	f shares outst	anding from
11 months ago	PROFIT	is profitabilit	y measured	as earnings	over book equ	uity. R1 is o	ne-month la	gged return.	R212 is the f_{412} for f_{412}
equity. TURN	is stock turn	t monuts pri tover measury	ed as trading	evious mont g volume ove	The state of slave of slave states and state	hares outstar	neu as une r iding.	niarket value	s mun nu n

Table 17: Multivariate Fama-MacBeth regression coefficients